Author's response to reviews

Title: The Effect of Proteoglycans Inhibited by RNA Interference on Metastatic Characters of Human Salivary Adnoid Cystic Carcinoma

Authors:

Hong Shi (shihong_hb@163.com)
Jie Wang (wangjiephd@163.com)
Fusheng Dong (dongfsh@126.com)
Xu Wang (wangxu63@hotmail.com)
Hexiang Li (Lihexiang-2008@163.com)
Yali Hou (yl_hstudent@sina.com)

Version: 10 Date: 12 December 2009

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

We have checked over our manuscript very carefully. There were some typographical changes and we have made the changes. We will submit a revised version today and confirm this is our final revised manuscript. Thank you for your time!

With best wishes,

HongShi